Author's response to reviews

Title: Effect of oral administration of ethanolic extract of Vitex negundo on thioacetamide-induced nephrotoxicity in rats

Authors:

Farkaad A Kadir (faalhadi@yahoo.com)
Normadiah M Kassim (normadiah_mk@um.edu.my)
Mahmood A Abdulla (mahmood955@yahoo.com)
Wageeh A Yehye (wdabdoub@yahoo.com)

Version: 3 Date: 18 October 2013

Author's response to reviews: see over
Dear Editorial Board,

First, we would like to admire reviewers and editorial efforts for the comments. We highly appreciated their time and attention to evaluate our work. Below, are our respective responses.

Best Regards,

Prof. Dr. Normadiah M. Kassim

Comments:

1. Thank you for including an ethics statement in your manuscript. However, we would ask you to please include the full name of the ethics committee that approved your study in this statement.

   The full name of ethic committee was added to the revised MS as shown in page 7.

2. While we appreciate that improvements have been made, we still feel that your manuscript would benefit from some further revisions to the grammar and quality of written English.

   The MS was sent to professional editing service for editing as shown in the revised MS

3. Please also ensure that your revised manuscript conforms to the journal style.

   The revised MS was re-written according to the journal style.

Thank you in advance your consideration,

Corresponding author,

Prof. Dr Normadiah M. Kassim